Author's response to reviews

**Title:** The successful introduction of an adapted form of the mini extra corporeal circulation used for cardiac surgery in an obese patient.

**Authors:**

Patrizio Sartini (patrizios@gmail.com)
Anna Winfield (annawinfield@doctor.org.uk)
Federico Bizzarri (federico.bizzarri@uniroma1.it)

**Version:** 2 **Date:** 21 January 2012

**Author's response to reviews:** see over
To the Editor of the Journal of Cardiothoracic Surgery

Dear Editor,

We are pleased to submit the reviewed copy of our paper entitled “The successful introduction of an adapted form of the mini extra corporeal circulation used for cardiac surgery in an obese patient”.

To this purpose I declare that I have read the comments of the reviewers and done the proper corrections: we took off the sentence in the abstract as asked by the reviewer 1 and improved the quality of figure 1 as stated by the reviewer 2.

I thank you in advance for the attention given to our paper.

Federico Bizzarri